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Subject: NBI Upgrade Weekly Status 5/21/09
Al, 

NBI Decon:  Preparations are underway and procedures are in review for the removal of BL4 Ion Dump which is tentatively scheduled for next week. The ion dump stand has been placed in the Test Cell near the calorimeter. The calorimeter has been surveyed to characterize the various levels of tritium contamination; thus far the levels are as expected. 

NBI Power and Control:  Layout work continues on power and controls systems. The cabling list has been finalized for power cabling. Cable lengths with various routings will be evaluated to look at line voltage drop and stored energy. Controls fiber optics requirements are being evaluated.

NBI Duct, VV Port, and Armor:  Extensive work has been done to develop the Bay K vacuum vessel modifications required for the wider beam tangency radii. A meeting was held to discuss some alternatives and then more modeling was done to capture some additional changes. Analysis work will now evaluate the stresses in this area. The additional changes have significant merit in that the beam free aperture is broadened across the interstitial area between Bay K and J eliminating concern about stray beam deposition in the throat of the duct. 

NBI BL Services: Layout work continues on services. Most of the service work has been finalized. Piping runs and quantities are being tallied for input to the cost estimate.

NTC Equipment Relocations: The NTC General Arrangement drawing is ongoing and has been updated with all received input to date.
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